

ovided by Repositorio Institucional de la Biblioteca del Banco de España

2015

SECURITIZATION AND BANKS' CAPITAL STRUCTURE

Andres Almazan, Alfredo Martín-Oliver and Jesús Saurina

Documentos de Trabajo N.º 1506

BANCODEESPAÑA

Eurosistema

SECURITIZATION AND BANKS' CAPITAL STRUCTURE

SECURITIZATION AND BANKS' CAPITAL STRUCTURE (*)

Andres Almazan

UNIVERSITY OF TEXAS AT AUSTIN

Alfredo Martín-Oliver

UNIVERSITAT ILLES BALEARS

Jesús Saurina

BANCO DE ESPAÑA

(*) This paper is the sole responsibility of its authors, and the views represented here do not necessarily reflect those of the Banco de España or the Eurosystem. We would like to thank E. Benmelech, C. Pérez-Montes, V. Salas and S. Titman for their valuable comments and suggestions. Any remaining errors are our own responsibility.

Documentos de Trabajo. N.º 1506 2015

The Working Paper Series seeks to disseminate original research in economics and finance. All papers have been anonymously refereed. By publishing these papers, the Banco de España aims to contribute to economic analysis and, in particular, to knowledge of the Spanish economy and its international environment.

The opinions and analyses in the Working Paper Series are the responsibility of the authors and, therefore, do not necessarily coincide with those of the Banco de España or the Eurosystem.

The Banco de España disseminates its main reports and most of its publications via the Internet at the following website: http://www.bde.es.

Reproduction for educational and non-commercial purposes is permitted provided that the source is acknowledged.

© BANCO DE ESPAÑA, Madrid, 2015

ISSN: 1579-8666 (on line)

Abstract

Asset securitization offers banks the possibility of altering their capital structures and the financial intermediation process. This study shows that the introduction of securitization is associated with fundamental changes in the funding policies of banks. In particular, we present evidence of more intense use of securitization by banks (i) with stronger growth opportunities; (ii) with liquidity constraints; (iii) with costlier alternative sources of funding; and (iv) with restricted access to capital markets owing to adverse selection. Securitization is also observed to be higher on the pecking order of financing choices of small and medium-sized banks and non-listed banks, which are likely to face more severe adverse selection problems.

Keywords: securitization, capital structure, adverse selection, pecking order.

JEL Classification: G32, G21.

Resumen

La titulización de activos ofrece a los bancos la posibilidad de alterar su estructura de capital y el proceso de intermediación financiera. Este trabajo muestra que la introducción de la titulización está asociada con cambios fundamentales en la política de captación de fondos de los bancos. En particular, presentamos evidencia de un uso más intenso de la titulización por parte de los bancos i) con mayores oportunidades de crecimiento, ii) con restricciones de liquidez, iii) con alternativas más costosas de financiación, y iv) con restricciones de acceso al mercado de capital debido a la selección adversa. Se observa también en este trabajo que la titulización ocupa un lugar importante en el orden de prelación en la elección de fuentes de financiación de los bancos medianos y pequeños y de los no cotizados, que son los que probablemente tienen problemas de selección adversa más severos.

Palabras clave: titulización, estructura de capital, selección adversa, orden de prelación en la financiación.

Códigos JEL: G32, G21.

1 Introduction

How does the introduction of securitization affect banks' operations and financing choices? Asset securitization expands the financing possibilities of banks by allowing the transformation of illiquid assets on bank balance sheets, e.g., mortgage loans, into marketable securities. Banks can use this financial innovation to fund asset growth and also to alter their capital structures, with securitizations substituting for more traditional liabilities such as bank deposits. Indeed, securitization is a financial innovation that fundamentally affects the financial intermediation process performed by banks and is reflected on the asset and liability sides of banks' balance sheets.

The financial literature has studied some determinants of the expansion of securitization, such as the corporate-taxation advantages in Pennacchi et al. (2014), and has also examined its effects on banks' operations, credit standards and credit expansion, e.g., Loutskina and Strahan (2009) and Loutskina (2011).¹ However, the literature has not studied in detail how securitization changes the financing choices and capital structure of banks, which is the focus of this paper.

Using data from the Spanish banking system during the period 1988-2006 we empirically examine a series of hypothesis on the effects that securitization may have produced on banks' capital structures. We motivate these hypotheses with insights from traditional theories of capital structures in corporate finance; namely the trade-off and the pecking order theories, (Myers 1984, and Myers and Majluf 1984). In particular, as suggested by the trade-off theory we assume that there is an optimal capital structure for each bank, which is potentially affected when access to a new financing possibility (securitization of assets in our case) becomes available. Based on this simple insight, we test hypotheses that consider whether securitization will be used more intensively by (i) those banks with more profitable uses for new funds and (ii) those banks for which access to other financing sources was restricted or particularly costly. Furthermore, motivated by the logic of the pecking order theory we examine the position securitization takes in the pecking order of bank financing choices and relate that to the adverse selection effects that banks may experience when issue securities.

The Spanish banking system provides an ideal framework for studying securitization as a shock on the availability of bank financing alternatives for several reasons. While in other countries, such as the United States, securitization developed progressively beginning in the early eighties; in Spain the process can be better described as a regime shift. Only after several legal changes that occurred in 1998 could banks effectively consider securitizing their assets.² The sample period ends in 2006, just before the financial crisis has removed loan securitization as a funding possibility for banks.³ In addition, securitization was promptly embraced by Spanish banks which, on average, proceeded to securitize a substantial part of

^{1.} See also Keys et al. (2010), Purnanandam (2011), and Demyanyk and Van Hemert (2011), who provide evidence on the relation between deterioration of loan quality and securitization. Other relevant references on this topic include Pais Rodriguez (2005, 2009), Mian and Sufi (2009) and Jiménez et al (2010). There is also evidence that securitization does not necessarily worsen adverse selection problems, as shown by Benmelech et al. (2012) for the securitization of corporate loans.

^{2.} The securitization period began in 1999, when the euro was adopted, facilitating firms' access to European capital markets. Bris, Koskinen and Nilsson (2009) show a reduction of firms' cost of capital after 1999.

^{3.} Since mid-2007, Spanish banks have only carried out securitization operations to obtain liquidity from the ECB. Private investors have refused to participate with new funds in this securitization market.

their assets (e.g., more than 25% of their mortgages).⁴ Finally, the Spanish banking system includes entities of different characteristics in terms of their access to finance, which provides a useful source of exogenous variation for our tests. In particular, this heterogeneity provides a useful framework to study the extent to which securitization might overcome adverse selection in capital markets.

Our study is organized into three main parts. In the first part, we provide a descriptive analysis of (i) the differences between the capital structure of Spanish banks before 1999 and at the end of 2006, (ii) the relation of securitization with the decoupling of lending and deposit activities of banks, and (iii) the changes in banks' funding policies and the role of securitization as a source of funds.⁵ We examine in particular the different roles of deposits in the Spanish credit expansions of 1988-1991 (pre-securitization period) and 1998-2006 (post-securitization period).

The second part of this study examines the determinants of securitization. Motivated by our premise that securitization affects the optimal mix of financing sources, we examine whether securitization is used more intensively by those banks with higher growth opportunities, higher financial costs of alternative sources of funds and lower costs of securitization. We consider several aspects of the securitization process including whether or not banks used securitization, but also the loan amounts securitized and time until the first use of securitization.

In the third part we examine how securitization fits in the pecking order of financing choices by banks. As bank balance sheets are opaque (Morgan, 2002), adverse selection can impose greater discounts on the debt and equity issuances of banks than on financial securities issued by nonfinancial firms. Furthermore, the sale of individual bank loans and pools of assets (with no tranching)⁶ can also be subject to large discounts, because banks have private information⁷ on borrowers' conditions (DeMarzo, 2005; DeMarzo and Duffie, 1999). In this context, banks might raise new funds at a lower cost through securitization because the process of pooling and tranching loans has the potential to reduce informational asymmetries (DeMarzo, 2005). To examine the securitization decision we modify the pecking order equation⁸ in Shyam-Sunder and Myers (1999) and Frank and Goyal (2003) and explore whether banks are likely to first choose securitization over other sources of funds and if this effect is more pronounced when they face more severe adverse selection problems (Bharath et al., 2009).

There are a number of findings that emerge from our analysis. First, loan securitization is associated with substantial changes in the liability structure of banks. In particular, the use of securitization is associated with lower reliance on deposits to finance the

^{4.} During the study period Spanish banks became the second largest issuers in Europe of ABS (after British banks) and of covered bonds (after German banks).

^{5.} From 2005 onwards, International Financial Reporting Standards (IFRS) forced Spanish banks to keep in their balance sheets securitized loans unless a *substantial* part of the risk and profits of these securitizations had been transferred. In practice, banks held more than 90% of their securitized loans on their balance sheets. We keep track of all the securitized loans. In order to homogenize the data and facilitate comparisons, we add back into bank balance sheets any securitized loan pool that was off-balance-sheet during the sample period.

^{6.} See the seminal article by Pennacchi (1988) on the process of selling loans and its associated risks. Drucker and Puri (2009) document substantial growth in the U.S. secondary loan market in spite of informational problems (a growth rate of 25% from 1991 to 2006 has led to a market size of \$236.6 billion in 2006).

^{7.} Adverse selection can be reduced by the presence of implicit agreements, as in Gorton and Pennacchi (1995), and restrictive covenants, as in Drucker and Puri (2009). In collateralized loan obligations (CLO), adverse selection can be reduced because securitized loans are fractions of syndicated loans, and reputation as well as the stake of lead's bank, reduce information asymmetries (Benmelech et al., 2012).

^{8.} Frank and Goyal (2008) provide a survey of the literature of the pecking order theory.

expansion of bank credit. These changes in the liability structure are more apparent for banks that face severe adverse selection problems and have restricted access to financial markets. Second, securitization is used more frequently by banks with stronger growth opportunities, higher relative cost of financing alternatives and tighter liquidity constraints. Bank size is positively associated with securitization, but large banks also make above-average use of equity and debt financing. In relative terms, securitization represents a higher proportion of external funds for smaller and medium-size banks. Securitization also tends to be higher in the pecking order of financing choices of small- and medium-size banks and non-listed banks, which are likely to face more severe adverse selection problems.

The existing literature on bank capital structure generally considers theoretical frameworks focused on the combination of lending and deposit activities.⁹ In this traditional banking setting, earlier articles have studied the impact of market conditions and asymmetric information on the optimal financial structure of banks, e. g., Gatev et al. (2009), Gatev and Strahan (2006), Diamond and Rajan (2001), Flannery (1994) and Calomiris and Kahn (1991). By contrast, this study focuses on a banking setting in which securitization is a relevant source of funds and new loans can be financed without the need of bank deposits, as found in Loutskina (2011). For this alternative banking model, we analyze how securitization affects the optimal asset composition and capital structure of banks. This analysis contributes to the literature exploring how securitization has changed traditional banking, e. g., Mian and Sufi (2009), Loutskina and Strahan (2009), Parlour and Plantin (2008) and Greenbaum (1987). Our work is also related to Cornett et al. (2011), who find a negative relation during the recent financial crisis between securitization and both bank holdings of liquid assets and bank lending. We find that securitizing banks expanded loans and reduced holdings of liquid assets more aggressively than other banks during the financial boom, indicating that the results in Cornett et al. (2011) are not only specific to the financial crisis.

The rest of the paper is organized as follows. Section 2 details the characteristics of the data used. Section 3 describes the effects of the introduction of securitization on banks' capital structures. Section 4 analyzes banks' securitization decisions with regression models. Section 5 examines the position of securitization in the hierarchy of financing sources of banks. Section 6 concludes the paper.

^{9.} Strahan (2008) provides a summary of the different banking theories that explain the rationale for the combination of lending and deposit activities inside the banking firm.

2 Data and Sample Characteristics

We collect data on securitized loan issuances of Spanish banks from 1999 to 2006, covering a period of active participation of Spanish banks in securitization activities. Before 1999, financial regulation limited the possibilities of Spanish banks to use loan securitizations. After 2006, the lack of liquidity in securitization markets prevented new issuances of securitized loans.

We use the term "bank" to refer to all forms of depository institutions, including (i) commercial banks, (ii) savings banks (i.e., "*cajas*") and (iii) credit cooperatives. Banks differ in their ownership structure, governance and organizational purpose. Commercial banks are forprofit corporations owned by their shareholders. Savings banks are nonprofit organizations controlled by local and regional governments. Credit cooperatives are entities owned by a fraction of their depositors and have as their main objective to provide credit to them. Historically, (i.e., before 1999) commercial banks raised external funds through different sources (i.e., issuance of equity and debt securities), rather than being restricted to use only bank deposits while, by contrast, savings banks and credit cooperatives were mostly limited to funding through bank deposits.

Our sample consists of the population of Spanish banks, which ranges from 212 banks in 1999 to 179 entities in 2006.¹⁰ In 1999 (2006), the sample includes 72 (51) commercial banks, 48 (45) savings banks and 92 (83) credit cooperatives. The reduction in the number of banks is due to mergers and consolidation processes.¹¹

Loan securitizations by Spanish banks include both issuances of securities backed by a particular portfolio of loans, which can be either mortgages (MBS or mortgage backed securities) or other types of loans¹² (ABS or asset backed securities), and issuances of covered bonds (*cédulas hipotecarias* in the original Spanish name). Issuances of MBS and ABS are initiated with the sale of a portfolio of loans from the originating bank to a special purpose vehicle (SPV), which then issues the MBS or ABS to investors in exchange of funds that are transferred to the originating bank. Typically, the originating bank also services the loan portfolio (i.e., receives the monthly payments, addresses arrears, etc.). Before 2005, regulation allowed banks to remove all the loans in MBS/ABS from their balance sheets, with the corresponding reduction in capital requirements. However, after 2005, a new accounting rule imposed on banks stricter requirements to remove securitized loans.¹³ Specifically, after 2005, Spanish banks can reduce regulatory capital requirements by issuing MBS/ABS only if these operations actually transfer credit risk out of their balance sheets. Regulatory recognition of risk transfers requires that banks do not provide SPVs with credit enhancements, which compensate investors in the event of losses in the securitized portfolio.

^{10.} We exclude branches of foreign banks, which have a negligible presence in retail banking in Spain.

^{11.} When banks merge, we consider them as separate entities before the merger and as a unique bank after the merger.

^{12.} Until 2004, banks only securitized mortgages (MBS) or issued covered bonds. From 2004 to 2006, MBS and securitization of covered bonds are still predominant, but banks started to issue ABS backed by auto loans, consumer loans, SME loans, loans to large firms and loans to the public sector.

^{13.} See the Appendix for more details on the change of requirements considered in the new regulation (Circular Banco de España CBE 4/2004).

A covered bond is secured not only by the full credit of the originating bank, but also by an eligible mortgage portfolio that acts as its specific collateral.¹⁴ Two regulatory requirements limit the issuance of covered bonds: (i) The eligible portfolio of collateralized loans can only include loans with a loan to value (LTV) less than 80%; and (ii) the amount securitized must be less than 80% of the value of the eligible portfolio (i.e., overcollateralization requirement). It is worth noting that the issuance of covered bonds has no immediate effect on regulatory capital. Eligible loans that back the covered bond remain in the originating bank's balance sheet and required regulatory capital stays constant.

From 2001 onwards, some groups of small banks securitize loans by first issuing covered bonds and then transferring those covered bonds to a joint SPV, which then issues bonds to investors. Small banks benefit from this multiple-bank securitization procedure, which improves the diversification of the underlying pool of assets and thus attracts additional investors. In our analysis, we consider the regular issuances of covered bonds and multiple-bank securitization as comparable in terms of the securitization decision, because they have similar economic and regulatory implications for the originating banks.¹⁵ The main difference between these two forms of securitization is the type of issuer: small and regional banks formed groups and used multiple-bank securitization, whereas larger banks with access to capital markets issued covered bonds directly. From an economic perspective, however, both mechanisms transform illiquid assets stocked in the balance sheet of banks into tradable securities.

We collect banks' financial and accounting information from the confidential statements reported regularly to Banco de España, who is the regulator and supervisor of the Spanish banking system. These statements include bank balance sheets, income statements and statements of regulatory capital collected at the end of each calendar year from 1999 to 2006. Additionally, we gather data on securitization issuances from two sources: (i) information for MBS/ABS is obtained from the brochures provided to investors by request of the Spanish financial market regulator (CNMV); (ii) information for covered bonds is obtained from incomplete partial set of brochures, which we complement with the balance sheet data in the confidential reports to Banco de España.

Table 1 shows the number of banks that securitize for every year and form of bank. Out of the 212 banks that exist at the start of the sample, 103 securitize loans at least once during the sample period. Table 1 also reveals that the amount of securitized loans increases substantially for all types of institutions (e.g., from 1999 to 2006, the amount of securitized loans increases approximately sixteen-fold for commercial and savings banks). The main issuers of securitizations in absolute volume are savings banks and commercial banks. Nonetheless, the securitization activity for credit cooperatives has been non-negligible, reaching a market share of 4.5% in 2006 which is similar to its weight in terms of total assets, 4.2%.

^{14.} This is similar to a secured bond issued by a non-financial corporation, which is guaranteed by both specific collateral and also the credit of the corporation itself.

^{15.} By 2006, multiple-bank securitization represented 41% of the total amount of covered bonds issued.

3 Securitization and financing choices

In this section, we describe the Spanish banks' financial condition during the period 1988-2006. We compare the banks' conditions in two sub-periods: a) the *pre-securitization* years (from 1988 to 1997) and b) the *post-securitization* years (from 1998 until 2006). As the issuance of securitized loans is viable on a large scale only after 1998, this comparison gives us a first approximation of the effects of securitization on bank behavior. In the post-securitization period, we also compare the banks that use securitization to obtain financing with the banks that do not securitize their loans.

3.1 Securitization and the evolution of balance sheets

We group bank balance sheet accounts in different subcategories. On the asset side, we consider three groups of items: (1) *LOANS*, which measures credit of all maturities granted by a bank to the non-financial sector (i.e., households and firms); (2) *GOVBONDS*, which accounts for the amount of government debt held by a bank; and (3) *INTERBANK*, which reflects a bank's net financial position in the interbank market (i.e., the difference between lent and borrowed funds, including the net position with the central bank). On the liability side, we consider four groups: (1) *OWNFUNDS*, which measures a bank's equity position (i.e., capital, reserves and insolvency funds); (2) *DEBT*, which corresponds to the amount of debt financing issued by a bank in wholesale markets (excluding the interbank market); (3) *DEPOSITS*, which includes traditional demand deposits held by banks and (4) *SEC*, which consists of the sum of securitized instruments issued by a bank. In addition to these items, we calculate a residual account, i.e., *REST*, which is computed as the difference of the assets and liabilities not considered in the other groups described in this paragraph.¹⁶

A number of stylized facts emerge from the aggregate data of the different groups of bank accounts, which is displayed in Table 2 and in Figure 1. On the asset side, the emergence of securitization is associated with an increase in the weight of *LOANS* with respect to total bank assets, which rises from 68% in 1997 to 84.6% in 2006. By contrast, this weight remains fairly stable in the pre-securitization years. The growing importance of *LOANS* in the balance sheet is achieved at the expense of government debt (*GOVBONDS*), whose weight relative to total bank assets is reduced from 17% in 1997 to 4.1% in 2006.

On the liability side, the start of the securitization period is connected with abrupt changes to the capital structure of Spanish banks. In the pre-securitization period, *SEC* is negligible, whereas *DEBT* and *OWNFUNDS* respectively represent an average of 5.1% and 10.8% of bank liabilities. During these years, bank deposits are the dominant form of bank financing, i.e., the group *DEPOSITS* represents 84.2% of bank liabilities. From 1998 onwards, there is a drastic reduction of bank deposits (*DEPOSITS* amounts to 59.1% of bank liabilities in 2006), an increased reliance on securitization (*SEC* represents 19.8% of bank liabilities in 2006) and, to some extent, on wholesale debt financing (i.e., the weight of *DEBT* with respect to bank liabilities rises to 12.3% in 2006). This reliance on market debt financing was a major shift in the capital structure of Spanish banks and it is analyzed in further detail in section 4.

^{16.} More specifically, *REST* includes the following asset side items: other holdings of financial assets (e. g, private fixedincome debt, cash, and derivatives), positions in the bank trading book, corrections for writing-off assets and other asset items. On the liability side, *REST* includes derivatives, other commercial obligations with suppliers, short positions in securities from overdrafts in repo operations and financial guarantees. For a more detailed explanation of all the groups and their components, see the Appendix.

Finally, the contribution of *OWNFUNDS* to bank liabilities is reduced to 8.7% in 2006, revealing a process of leveraging in the post-securitization period that has been also documented in previous studies.

Additional insight is gained from the comparison of the behavior of the banks that issue securitized loans to fund their operations and the banks that do not securitize. Figure 2 shows that the expansion of credit during the post-securitization period is particularly intense for banks that choose to securitize (Figure 2A). These banks increase the weight of *LOANS* relative to total assets by 17.4 percentage points during the post-securitize (Figure 2B), there is however no significant increase in their loan base. While the weight of *LOANS* relative to total assets for banks that do not securitize from 75.1% in 1997 to 79.2% in 2006, this variation is not statistically different from zero. In addition, the depletion of the stock of liquid assets (i.e., government debt) is larger for securitizing banks (a decrease from 17.3% in 1997 to 4.1%, significant at 1%) than for those banks that do not resort to securitization (a decrease from 13.5% to 5.2%, significant at 5%).

On the liability side, there are also significant differences between securitizing and non-securitizing banks. Most notably, the reduction of the weight of *DEPOSITS* with respect to total liabilities is particularly intense for securitizing banks, with a decrease from 84.1% in 1997 to 58.6% in 2006 (banks that do not securitize observe a change from 76.9% in 1997 to 73.5% in 2006). This difference occurs because securitizing banks present an average deposit growth rate of 10.6%, which is inferior to the growth rate of 13.6% for the deposits of non-securitizing banks. In addition, securitizing banks use the funds obtained from securitization to substitute for deposits as a source of funds.¹⁷

3.2 Securitization and the reliance on deposits for credit expansion

Our previous findings suggest that securitization contributes to the decoupling of the deposit and credit activities of financial intermediaries. To further examine this issue, we analyze the relation between credit and deposits in two periods of intense economic expansion in the Spanish economy: (i) the period 1988-1991, when securitization was unfeasible, and (ii) the period 2003-2006, when securitization was fully operative. In the former period, loan growth was limited by the evolution of deposits. In contrast, loans were able to grow in the latter period at a higher pace than deposits because banks had securitization as an additional financing source. It is precisely during an economic expansion when the demand for credit can increase at a higher pace than the volume of deposits.¹⁸ An excessive reliance on deposit funding can left unattended part of the potential demand for credit and preclude an efficient intermediation process.

As Figure 3 indicates, credit growth is more than two times the deposit growth rate in the 2003-2006 period, whereas it follows closely the rate of growth of deposits during the 1988-1991 period. The higher credit growth during the post-securitization period is likely due to a positive effect of securitization on bank credit supply rather than to a higher demand

^{17.} Non-securitizing banks relied on debt issuances, which reached up to 15.6% of their assets, to fund their loan expansion. Securitizing banks also issued debt (12.2%), but used securitization more intensely (20.5%).

^{18.} Cornett et al. (2011) argue that the volume of core deposits might decrease during expansion periods because savers look for higher returns, and it might increase during recessions because deposits are perceived as safe assets. Since Table 2 shows that the volume of total deposits in Spain increases over time, our finding suggests that the recourse to securitization is not just compensating a fall in the volume of deposits, but it can be also due to an expansion of the demand for loans higher than deposit growth.

derived from stronger economic growth. In fact, the average GDP growth rate in the postsecuritization period was lower (3.5%) than in the pre-securitization period (4.1%).

We examine more formally the link between deposit and credit growth across individual banks with a set of regressions of credit growth on deposit and GDP growth. Results are reported in Table 3. We consider both OLS and fixed effect specifications and both indicate that the coefficient of deposit growth falls by 40% for the total sample of banks from 1988-1991 to 2003-2006 (from 0.48 to 0.29 in OLS specifications (1) and (3), and from 0.35 to 0.19 in the fixed effects specifications in (2) and (4)). This result shows that the relationship between deposit and credit growth is less intense after banks can securitize, and it is also consistent with the hypothesis that securitization contributes to decouple the credit and deposit activities of banks (Loutskina, 2011; Loutskina and Strahan, 2009).

In Columns (5) to (8) of Table 3, we examine the relation between credit and deposit growth for the subgroups of securitizing and non-securitizing banks. In this analysis, the evidence is less conclusive. Securitizing banks exhibit a relatively low coefficient on deposit growth in the OLS specifications (i.e., 0.30 in (5) relative to 0.47 for non-securitizing banks in (7)). However, we fail to find a stronger relationship between credit and deposit growth for non-securitizing banks once we include fixed effects (i.e., the deposit growth coefficient is 0.27 for securitizing banks in (6) and 0.23 for non-securitizing banks in (8)). A possible reason for these mixed results is that some common factors explain both securitization decisions and the growth rates of loans and deposits, making more difficult the interpretation of coefficients in these simple regressions. To consider this issue and other possibilities, we examine more carefully next the factors that affect banks' securitization decisions.

4 The determinants of the securitization decision

We consider insights that emanate from existing theories of capital structure to the case of banking firms in order to derive some testable predictions about their decision to securitize. In particular, we hypothesize that securitization alters the optimal mix of banks' financing choices because it (i) provides a novel source of finance whose attractiveness depends on banks financial choices (*trade-off* theory) and (ii) facilitates the access to market financing of banks facing severe adverse selection problems (*pecking order* theory).

The *trade-off* theory insights are based on the premise that banks that alter their capital structure and decide to securitize are those banks that are financially constrained i.e., that find it relatively more costly to finance investment opportunities with pre-existing sources of funds. In order to identify financially constrained banks, we consider three different dimensions: (i) the relative cost of securitization with respect to other funding alternatives, (ii) the growth opportunities in the banks' loan portfolios, and (iii) the liquidity position of banks. Our hypothesis is that banks whose sources of funds are relatively more expensive, banks with more growth opportunities and banks with tighter liquidity conditions are more likely to benefit from the new financing possibility offered by securitization.

The *pecking order* theory suggests a more specific insight: securitization grants banks access to public market financing without facing an adverse selection high discount. More specifically, securitization mitigates adverse selection since it reduces the informational problems associated with the direct sale of loans or pools of loans (DeMarzo, 2005). To examine the empirical relevance of this insight, we analyze whether banks that are more affected by adverse selection costs are those that resort relatively more to raising market funds through securitization.

The rest of this section presents the variables that are used to proxy for the determinants of securitization among Spanish banks, an explanation of the empirical methodology employed for analysis and the results.

4.1 Variables

We distinguish five groups of explanatory variables: (1) proxies related to financing costs, (2) proxies related to liquidity, (3) proxies related to the growth opportunities of banks, (4) variables related to the market access possibilities of banks and (5) other control variables.

4.1.1 FINANCING COSTS

We have argued that the financial benefits of securitization are likely to be larger for banks that are constrained in their investment policy by their inability to resort to other sources of financing such as demand deposits, interbank loans and debt and equity issuances. These constraints could appear in the form of high financing costs associated with available funding alternatives. Banks can benefit from securitization provided that this financing source presents a lower relative financing cost than the existing alternatives.

Since we do not have price information to measure the costs of alternative financing sources, we use the relative amount of financial instruments held by the banks in the presecuritization period as proxies for the cost of the alternative financing sources. Implicitly this assumes that the observed capital structures of banks are the result of profit-maximizing strategies that take into account differences in the marginal financial costs of alternative financial sources. For instance, if a bank concentrates its financing in the pre-securitization period in only one source (i.e., deposits), one would expect that this source is particularly inexpensive for the bank (or alternatively that the financing cost of other funding alternatives is relatively higher). In this case, securitization would be less beneficial for banks with highly-concentrated sources of finance, (since this probably reflects the access to a low cost financing source) and would be more beneficial to banks whose capital structure contains a wide set of sources.

In particular, for each bank-year, we consider the following variables for relative financing costs:

(i) *Dep/Loans, Interbank/ Loans, Debt/Loans, Equity/Loans:* The ratio of each financing source with respect to total loans provides a measure of the constraints faced by a bank in its credit operations. We consider four possible sources of financing for banks: deposits, net financing from the interbank market,¹⁹ debt and equity. Banks that have better access to one of these funding sources (i.e., low financial costs, better availability of funds, an extensive branch network to access retail deposit markets) will finance a higher proportion of their loan operations with this financing source. We expect that these banks have lower incentives to securitize, because they already have access to a relatively cheap financing source.

(ii) Concentration: As an alternative measure of the concentration of financing sources we construct the ratio of the sum of squares of financing sources divided by the square of the sum of all the sources, that is, $\frac{Interbank^2 + Debt^2 + Equity^2}{(Interbank + Debt + Equity)^2}$. Concentration is

bounded between 1, when a bank has only one source of financing in addition to deposits, and 1/3, when a bank uses the same amount of the three market sources of funds. According to the logic stated above, banks with a dominant financing source will feature a higher *Concentration* measure and should exhibit a lower tendency to securitize. It is important to note that we do not include deposits in this ratio and consider them in the separate variable *Dep/Loans* to isolate the effect of this traditional and large source of bank financing from financing alternatives in public financial markets.

4.1.2 LIQUIDITY

Banks with have higher (lower) liquidity constraints are potentially subject to higher investment constraints and, thus, they are more likely to securitize. We use banks liquidity holdings as an inverse measure of their liquidity constraints. Specifically, we consider two variables to describe the liquidity of a bank:

(i) Liquidity / Loans: This numerator of this measure consists of the sum of the government debt and the net volume of deposits held in the interbank market, including the net position with the central bank. This definition of liquid assets is similar to the measure suggested in Basel III, i.e., High Quality Liquid Assets in the Liquidity Coverage Ratio, LCR. We expect a negative relation between a bank's incentives to securitize and the ratio of this liquidity buffer with respect to the volume of loans that are to be financed.

^{19.} The net interbank position is defined as Max{Interbank Loans - Interbank Deposits, 0}.

(ii) Past profitability / Loans: This variable is a proxy of the availability of internally generated funds that can be used to finance new loans. This variable is computed as the ratio of the profits of the previous year net of the distributed dividends with respect to the volume of loans to be financed. We expect that banks with higher retained earnings will have lower incentives to securitize.

4.1.3 GROWTH OPPORTUNITIES

Banks with potentially higher growth opportunities are more likely to need financing, and securitization may help to cover this need. In order to capture growth potential, the existing literature suggests the use of the price-to-book ratio (e.g. the Tobin's Q ratio)²⁰ however, since a very large part of the banks in our sample are non-publicly traded we consider the following alternative proxies:²¹

(i) Projected Loan Growth: We estimate a series of projected loan growth (which we identify with banks growth opportunities) that we define as: $\frac{L_t - L_{t-1}}{A_{t-1}}$, where L_t is the

balance of loans at the end of year t, and A_{t-1} is the total assets at the end of year t-1. We use the ratio of the absolute difference of loan balances on year t and t-1 with respect to total assets on year t-1 to avoid large growth rates caused by small initial loan balances and to be consistent with the rest of variables defined below. Specifically, we estimate an autoregressive model that explains loan growth on year t as a function of the loan growth on year t-1 and t-2with a rolling window of 10 years, which avoids differences in standard errors due to the growing number of years for observations that are observed later in the time horizon. For each year after t, we obtain the best prediction (based on the observed loan growth for years t-1 and t-2) of loan growth. The variable loans, L_t , includes loans to the public sector and loans to non-financial firms and households (both residents and non-residents).

We validate this proxy (i.e., the extent to which it measures bank growth opportunities) by considering its statistical relationship with the Tobin's Q ratio for the subsample of banks whose equity is publicly traded. In particular we regress Tobin's Q using as explanatory variables *Projected Loan Growth* and the rest of the explanatory variables considered in our model of the decision to securitize. We find that *Projected Loan Growth* is statistically significant at 1% while the rest of variables are not significant (OLS with robust standard errors clustered at bank level) and the R² amounts to 23%. When *Projected Loan Growth* is the only explanatory variable, the sign, magnitude and statistical significance of the coefficient remains unchanged, and it can explain up to 12% of the total variation of the price-to-book ratio.

(ii) As an alternative proxy for growth opportunities we use *GrowthOpp* i.e., the number of new regional markets in which banks enter to operate and the sum of GDP for the regional markets of bank operations. This alternative proxy is considered in the robustness analysis to test the validity of the results of the main specifications.

4.1.4 MARKET ACCESS

Banks' accessibility to financial markets can affect their use of securitization in different ways. On the one hand, a bank with previous access to the market can use this new financing

^{20.} See Frank and Goyal (2008) for a review.

^{21.} Savings banks and credit cooperatives are not listed in the Stock Market because of their legal nature. As for commercial banks, only 14 banks out of 51 in our sample were listed in the Stock Market.

channel without a big investment in market recognition and lower transaction costs. On the other hand, securitization opens radically new financing possibilities for private banks which have been excluded from market access in the past. To analyze these alternatives we consider two different proxies:

InAssets: Larger banks are more likely to encounter lower transaction costs of accessing financial markets and, thus, they be more likely to securitize.

Savings and Coop: Savings banks and credit cooperatives had very restricted access to financial markets before securitization. Therefore, they are more likely to find securitization attractive. These two dummy variables take the value of 1 if the bank is a savings bank or a credit cooperative and zero otherwise. We expect a positive coefficient if the hypothesis that securitization enables firms to reduce the costs of adverse selection holds in the data.

4.1.5 OTHER CONTROL VARIABLES

These variables aim to capture whether the decision of banks to securitize has been driven by other securitization determinants, such as the possibility to manage credit risk in their portfolios or to perform regulatory capital arbitrage. We consider three proxies:

(i) *NPL*: The ratio of non-performing loans over total loans (*NPL*) in bank portfolios reveals their credit standards and levels of accumulated risk. We expect banks with a higher *NPL* to have riskier loans and stronger incentives to transfer those risks to investors via securitization.²²

(ii) *RegCap*: A dummy variable that takes the value of 1 if the Basel regulatory capital ratio is below the 25th percentile of the distribution and zero otherwise.²³ The regulatory capital ratio is computed by dividing regulatory capital (the portion of capital eligible to fulfill the capital requirements of the Basel Committee) by bank assets weighted according to their risk (Risk Weighted Assets or *RWA*). Banks closer to the regulatory limit, which is set at 8% in the Basel requirements, can find useful to use ABS/MBS as an instrument to ensure regulatory compliance.

(iii) *Mortg/Loans*: Mortgages are the most common underlying asset used in securitizations. Therefore, the weight of mortgage loans in a bank balance sheet controls for the possibility that banks with a higher proportion of mortgages securitize more often.

4.2 Empirical strategy and results

We perform three sets of tests. First, we estimate two alternative specifications of a Probit model to investigate the determinants of the banks' decision to securitize (i.e., the "extensive margin"). We examine both the year-to-year decisions to securitize (with a panel dataset of bank conditions) and the decision to securitize at least once during the post-securitization period (with a cross sectional dataset of the initial conditions of banks at the start of this period). We consider this second estimation approach to account for the possibility that the decision to securitize is part of a long-term financial strategy and then require several years for its full implementation. Second, we estimate a Tobit model to analyze the determinants of the

^{22.} If riskier loans require more bank monitoring, this opposite effect may reduce the incentives to securitize high risk loans.

^{23.} As discussed below, we consider alternative definitions of this variable including other cut-off values.

amount of assets securitized by banks (i.e., the "intensive margin"), also considering the two estimation approaches used for the Probit model. Finally, we estimate duration models for the decision to securitize and explore which variables determine the speed at which a bank decides to securitize for the first time.

4.2.1 RESULTS ON THE DECISION TO SECURITIZE

Table 4 presents the marginal effects of the Probit regressions, which are estimated with robust standard errors clustered at the bank level. Columns (1) and (2) display results for the Probit models that relate the decision of securitizing at least once during the 1999-2006 period to the values on year 1999 of the proxy variables for the determinants of securitization. The year 1999 is the first when securitization became available to banks. Column (1) includes all the financing alternatives relative to the volume of loans (*Dep/Loans, Interbank/Loans, Debt/Loans, Equity/Loans*) and column (2) replaces these variables with *Concentration*. As shown in (1) while other financial cost proxies are not statistically significant, the coefficient of *Dep/Loans* is negative and statistically significant in (1) which suggests that banks with lower relative marginal cost for deposits at the beginning of the securitization year are less likely to securitize during that period. In (2) *Concentration* has the expected negative sign but fails to be statistically significant.

Banks with a higher stock of liquid assets with respect to loans (*Liquidity/Loans*) in 1999 are less likely to securitize. However, we do not find evidence that internally generated funds from past profits (*Past Profit/Loans*) reduce the incentives to securitize.

The coefficient on *Projected Loan Growth* is positive and highly significant which indicates that banks used securitization to fund credit growth, in line with preliminary evidence in Figure 3. Securitization is associated with a substantial increase in the growth rate of banks' balance sheets, which increased from 8.9% per annum during the *pre-securitization* years to 14.0% during the *post-securitization* period.

There is also a positive and significant coefficient for InAssets, suggesting that large banks with an existing record of operations in financial markets can gain access to securitization. In addition, we find evidence that, other things equal, savings banks and credit cooperatives are more likely to securitize than commercial banks. This result is consistent with the hypothesis that securitization can reduce adverse selection problems if groups of banks can jointly issue bonds backed by a common loan portfolio.

As for other control variables, there is no evidence that Spanish banks used securitization as a risk management tool (i.e., to mitigate and transfer credit risk) or as a mean to engage in regulatory capital arbitrage. This result could be explained by the tougher regulation on conduits and SIV (Structured Investment Vehicles) in Spain, which was precisely designed to limit off-balance sheet risks and regulatory capital arbitrage.²⁴

Columns (3) and (4) provide results for the Probit models that relate the decision of securitizing at least once during the period 1999-2002 to the situation of the banks in 1999. The results are similar to the estimates in (1) and (2), suggesting that the banks that securitized during the period 1999-2006 already made the decision to securitize during the initial sub-period 1999-2002.

^{24.} Acharya, Schnabl and Suarez (2013) study conduits as a case of "regulatory arbitrage", and they notice that banks based in Spain and Portugal did not set up conduits.

Columns (5) and (6) show the estimates obtained with the panel data of year to year securitization decisions. In these models, the bank decision to securitize in year *t* is explained by the value of explanatory variables in year *t*-1. The qualitative results are similar to the previous estimations in columns (1) to (4), though the magnitude of the coefficients is generally smaller. The variable *Interbank/Loans* becomes statistically significant, but *Dep/Loans* loses its significance and none of the other financial cost proxies is significant in (5). As stated above, these weaker findings could be the result of the time frame for the decision to securitize. Securitization choices would not be made on a yearly basis, but they would rather form part of medium to long-term financial strategy.

We perform a number of tests to evaluate the robustness of the above results. First, we consider additional growth proxies as explanatory variables. Therefore, as alternative controls of bank growth possibilities, we include: (i) the (weighted) GDP growth of all the regions where a bank operates and (ii) a dummy variable that identifies banks opening branches in new regional markets. We expect that banks will have higher growth opportunities if they operate in regions with high economic growth or they have entered new markets. The coefficient for weighted GDP growth is significant at the 5% level if this variable is the only control for growth opportunities in the model for the decision to securitize. If both weighted GDP growth and the dummy identifying new entrants are included, the coefficient on the former variable is only significant at the 10% level. The dummy identifying new entrants is not significant even when included as the single proxy for growth possibilities. When we include these two proxies together with Projected Loan Growth, the latter variable is significant at the 5% level whereas the weighted GDP growth and the dummy for new entrants are found to be not significant. Thus, we interpret that the three variables are capturing a common effect, and that Projected Loan Growth captures all the relevant information contained in the other two variables.

For the panel data models, we also consider two additional dummy variables as proxies for access to financial markets: (i) a dummy for banks listed in the stock market and (ii) a dummy for banks that have issued at some point in time debt instruments in wholesale markets. The results show positive and significant coefficients for these variables, in line with the predictions of theoretical models. Nevertheless, these variables are not included in the main analysis because they are highly correlated with the identity of some banks, creating an overfitting problem (i. e., perfect predictions for some banks) in the regressions for the decision to securitize.

4.2.2 RESULTS ON THE AMOUNT SECURITIZED

Table 5 displays results for the models of the asset amounts securitized by Spanish banks (i.e., the "intensive margin"), which are estimated with a Tobit specification with standard errors robust to heteroskedasticity and clustering at bank level. The dependent variable is the amount of funds securitized by a bank normalized by the size of its assets, but the results presented in Table 5 still follow the same structure as the estimates of the Probit models in Table 4. The explanatory variables are the same as in the Probit analysis of subsection 4.2.1. For columns (1) and (2), the dependent variable is the amount of funds securitized by a bank during the whole 1999-2006 period (normalized by the size of its assets in 2006) and explanatory variables are fixed at their values on year 1999. Columns (3) and (4) present results for specifications analogous to (1) and (2), but with a shorter time horizon from 1999 to 2002. Columns (5) and (6) correspond to the results from panel data estimation, which uses the amount securitized on year *t* as dependent variable and the proxies on year *t*-1 as explanatory variables.

Comparing the coefficients of the explanatory variables on Table 4 and Table 5, we observe that the proxies for liquidity, growth opportunities and market access maintain their sign and statistical significance. These determinants of the securitization decision have a comparable qualitative impact on the intensive and extensive margins.

The conclusions about the effect of financial costs are also comparable in Table 4 and Table 5, but not all the coefficients for this sub-set of variables are equal across specifications. In Table 5, evidence of the effect of financial costs on securitization is supported by the negative and statistically significant coefficients of *Interbank/Loans* and *Equity/Loans* in columns (1) and (3) (instead of *Dep/Loans* as in Table 4) and *Dep/Loans* and *Interbank/Loans* in column (5) (instead of only *Interbank/Loans* as in Table 4). Finally, other control variables do not have a significant effect on the amount securitized, as it was the case for the decision to securitize.

4.2.3 RESULTS FROM DURATION ANALYSIS

Table 6 presents estimates of the duration models, which assume that the time elapsed until a bank securitizes for the first time is governed by a parameterized hazard rate. The function h(t, X) for the hazard rate is defined as the product of a common time factor and a function of bank level variables that captures observed heterogeneity across banks, i.e., $h(t, X) = h_0(t) \cdot e^{X_i \beta}$. Columns (1) and (2) in Table 6 report results for an exponential model with a constant conditional probability of securitization over time, $h(t, X) = e^{X_i \beta}$, while estimates in columns (3) and (4) correspond to a Weibull model. This alternative specification assumes a monotonic dependence of the hazard rate on time, $h(t, X) = pt^{p-1}e^{X_i\beta}$. The probability to securitize increases (decreases) over time if p>1 (p<1), whereas p=1 implies the reversion to the base exponential model. The results are presented in the form of exponential coefficients (i.e., e^{β}), which can be directly interpreted as increases in the baseline hazard rate.²⁵

When we allow for time dependency of the hazard rate, we observe that the probability to securitize increases over time: the estimates of p are 1.9 and 1.91 in the Weibull specifications in columns (3) and (4). This result is consistent with the increasing number of securitizations over time reported in Table 1. The sign and magnitude of the coefficients of bank level variables are not significantly affected by the assumption of time dependence of the hazard rate. We thus provide unified comments on these coefficients for the specifications with and without time dependence.

We observe that a higher deposit base (*Dep/Loans*) is associated with a longer time until first securitization, which is an outcome consistent with the results for the Probit and Tobit models. The coefficients on the *Concentration* variable in columns (2) and (4) are positive and similar in size, though we only observe a significant result in the Weibull specification. A high value of *Concentration* or *Dep/Loans* implies strong reliance on a single financing source during the pre-securitization period and, given the estimated coefficients, a longer period of time until the first securitization. These results for *Dep/Loans* and *Concentration* support again the hypothesis that a large base of a pre-existing financing

^{25.} For instance, $e^{\beta_{K}} = 1.2$ implies that an increase in 1 unit in X_{k} leads to an increase of the baseline hazard rate by 1.2 and a decrease in the expected time to securitize. On the contrary, $e^{\beta_{K}} < 1$ implies that an increase in 1 unit in X_{k} lengthens the amount of time until the bank securitizes.

source is associated with a low relative financial cost of that source and therefore a lower probability of using new financing alternatives.

Other variables with significant coefficients include *Projected Loan Growth, Savings, Coop* and In*Assets*. All of these variables receive coefficients higher than 1 and an increase in the value of one of them reduces the expected time until the first securitization. This result is consistent with the positive relation found between these variables and the decision to securitize, as reported in Table 4 and Table 5.

In order to evaluate the predictive power of the duration models, we present in Figure 4 the distribution across banks of the predicted number of years until first securitization. The specification in Column (3) of Table 6 is used to produce the predictions. We present separately the distributions of securitizing and non-securitizing banks. The model predicts a lower number of years until first securitization for banks that actually securitized. The distribution for this type of banks is concentrated around values below 5 years: 63% and 93% of the cases for securitizing banks, the distribution is more dispersed and the time until first securitization is predicted larger than 10 years for 78% of the cases, which is out of the temporal scope of the sample.

Empirical analysis based on different models and estimation techniques provides evidence consistent with some key theoretical predictions. More specifically, banks are found to be more likely to securitize when the costs of alternative sources of funds are higher, growth opportunities are higher and the proportion of liquid assets held in balance sheets is lower. There is also evidence that savings banks and credit cooperatives are more likely to make early use of securitization than commercial banks. We argue that the reason for this latter result is that securitization grants these banks with a form of access to financial markets that was not previously available to them. In the next section, we provide further analysis to of the link between the access to financial markets through securitization and the reduction of costs related to asymmetric information.

5 Securitization and the pecking order of financing choices

In this section, we further investigate the hypothesis of whether securitization particularly affects banks that bear high informational costs to access capital markets. Specifically, we examine how securitization fits in the pecking order of financing choices by banks. To examine the securitization decision we build on the analysis of Shyam-Sunder and Myers (1999) and Frank and Goyal (2003) and Bharath et al. (2009) and consider whether banks are likely to first choose securitization over other sources of funds and whether this effect is more pronounced when they face more severe adverse selection problems.

Our analysis modifies the conventional pecking order equation in previous empirical corporate finance to adapt it to the case of bank securitization. In a typical corporate finance case, the basic test examines whether a firm's financial deficit (*FD*) can explain the increase of its debt (ΔD).²⁶ More specifically, the typical study is to estimate a regression of the form: $\Delta D_{it} = \alpha + \beta FD_{it} + e_{it}$ and tests the hypothesis β =1 (i. e., financial needs are covered only by issuing new debt).

Similarly we estimate a regression of the amount of (new) securitized loans on the bank's financial deficit (*FD*). For bank *i* on year *t*, we define its financial deficit as:

$$FD_{it} = \Delta LOANS_{it} - \Delta DEPOSITS_{it} + \Delta INTERBANK_{it} + \Delta GOVBONDS_{it} + \Delta RESERVES_{it} + \Delta REST_{it}$$
[1]

where *RESERVES* consists of banks' reserves, including current profits, and the rest of variables have been defined in section 3. We then consider the following specification:

$$SEC_{it} = \alpha + \beta FD_{it} + e_{it}$$
^[2]

and estimate β for different types of banks.

Following the insights in Bharath, Pasquariello and Wu (2009) we argue that firms are more affected by pecking order considerations at the time of issuing new securities if adverse selection problems are severe. We also follow their specific methodology and include in the pecking order equation the interaction of *FD* with a measure of asymmetric information. Specifically, we interact the variable *FD* with three indicator variables: *Small, Savings* and *Coop*, which identify respectively small banks, savings banks and credit cooperatives. We define a bank as small if its total assets fall below the 30th percentile of the sample distribution of bank assets. In particular we estimate:

$$SEC_{it} = \alpha_0 + \alpha_1 Small_{it} + \alpha_2 Savings_i + \alpha_3 Coop_i + \alpha_4 Savings_i \cdot Small_{it} + \alpha_5 Coop_i \cdot Small_{it} + \beta FD_{it} + \gamma FD_{it} \cdot Small_{it} + \delta_1 FD_{it} \cdot Savings_i + \delta_2 FD_{it} \cdot Coop_i + e_{it}$$

$$[3]$$

and test the hypotheses: $\gamma > 0$, $\delta_1 > 0$, and $\delta_2 > 0$. Intuitively, the coefficient on *FD* is expected to be higher for small and non-listed banks, which are thought to finance a higher proportion of their financial deficit through the issuance of securitized loans.

^{26.} The previous literature usually rejects the null hypothesis of β =1 (Shyam-Sunder and Myers, 1999; Frank and Goyal, 2003; Fama and French, 2005) and finds values estimates of beta smaller than 1. Empirical evidence shows that firms typically combine the issuance of debt and capital to finance their financial deficits.

Table 7 presents our estimates for the sample of banks with a positive financial deficit. We report robust standard errors that are corrected for clustering at the bank level. The first column of Table 7 shows the results of the base specification in [2]. We find that β <1, which implies the rejection of a strict (and more narrow) version of the pecking order hypothesis. Column (2) in Table 7 presents the results for a specification that includes the interactions of FD with the indicators Savings and Coop. For credit cooperatives, we obtain a positive coefficient on the interaction with FD, which is significant at the 5% level. Specification in column (3) includes the interaction FD-Small, which also has a positive and significant coefficient. To disentangle the effects of size and legal nature, the specification in column (4) includes all the previously used variables and additional interactions of FD·Small with the dummies for Savings and Coop. The results show that FD, FD. Coop and FD-Savings-Small receive statistically significant coefficients. These estimates suggest stronger preference for securitization among medium-large credit cooperatives and small savings banks. Indeed, credit cooperatives and small savings banks are candidates to face severe informational problems and restricted access to debt and equity markets. In summary, Table 7 provides suggestive evidence that this type of banks used securitization as a tool to reduce adverse selection costs and raise funds in wholesale financial markets.

As a robustness check, we use alternative thresholds for the definition of small banks. Thus, we define a bank as small if its size is smaller than the 5th, 10th, 20th, 40th and 50th percentiles of the distribution of the banks' assets, as alternative limits to the 30th percentile used in Table 7. Estimation of specification (4) in Table 7 is then repeated with the redefined *Small* indicators. The results on the coefficients of *FD* and *FD*.*Coop* are not sensitive to the definition of *Small*. However, the coefficient of *FD*.*Savings*.*Small* is not significant if the size threshold used for *Small* is lower than the 15th percentile or higher than the 50th percentile.

6 Conclusions

Securitization enables banks to shift from a traditional business model financed through the expansion of deposits to new business models that rely on the demand for securitized assets in international financial markets. In this study we find that securitization is also associated with substantial changes in the capital structure and funding policies of banks, in particular a substantial reduction in their reliance on deposits as a financing source. In addition, securitizing banks are observed to increase their loan volumes and their ratios of loans over total assets, whereas they decrease their holdings of low-return liquid assets.

The transition to a more market-dependent financing model has enabled banks to decouple the evolution of their credit activity from the capacity to raise deposits. This financial development, however, has also increased the vulnerability of banks to shocks in the financial markets. Non-securitizing banks have maintained a high weight of deposits over total funds and they have been able to cope better with the higher financial constraints observed after the start of the crisis in 2007 (Cornett et al., 2011). In this study, we fail to find evidence that relates securitization to banks credit risk management or regulatory capital arbitrage. While this may be due to our particular sample, we are able by contrast to provide evidence that relates the banks' securitization choice to other financing choices made by banks in order to fund their credit expansions.

Our cross-sectional findings are also interesting and consistent with the idea that the benefits of securitization can be associated to certain bank characteristics that correlate with their capital structure choices. Our empirical analysis indicates a positive relation between securitization benefits and the presence of funding constraints on credit growth, which can take the form of high cost of financial alternatives or outright exclusion from financial markets. We find evidence that securitization can grant access to financial markets to banks that were previously excluded due to severe adverse selection problems. Securitization is higher in the pecking order of financing choices for these financially constrained banks than for banks less affected by information asymmetries.

During recent years, securitization has been stigmatized because of its relation with the financial crisis initiated in 2007. The number of new issuances and the total volume of securitized loans have both drastically decreased, transforming banks' capital structures. Banks have been forced to either revert to the traditional deposit-based model or to rely on funds from central banks to fill in the funding gap left by the decrease in securitization activity. However, there are perspectives of a medium to long-term recovery in securitization markets (Bloommestein et al., 2011). The results in the current article suggest that there can be consequences associated to the elimination of securitization as a source of bank financing. In particular, if credit expansion by banks is a desirable policy objective, it may be necessary to provide alternative and flexible financing sources to banks, particularly to those institutions which find it more difficult issuing capital throughout the traditional forms of finance i.e., debt and equity.

REFERENCES

ACHARYA, V.V., SCHNABL, P. and SUAREZ, G. "Securitization without risk transfer" Journal of Financial Economics, 107, 2013, 515-536.

BENMELECH, E., DLUGOSZ, J. and IVASHINA, V. "Securitization without adverse selection: The case of CLOs". Journal of Financial Economics, 106, 2012, 91-113.

BHARATH, S.T., PASQUARIELLO, P. and WU, G. "Does asymmetric information drive capital structure decisions?", Review of Financial Studies, 22, 2009, 3211-3243.

- BLOOMMESTEIN, H.J., KESKINLER, A. and LUCAS, C. "Outlook for the securitization market" OECD Journal: Financial Market Trends, 1, 2011, 1-18,
- BRIS, A., KOSKINEN, Y. and NILSSON, M. "The Euro and corporate valuations". Review of Financial Studies, 22, 2009, 3171-3209

CALOMIRIS, CH., and KAHN, CH. "The role of demandable debt in structuring optimal banking arrengements". American Economic Review, 81, 1991, 497-513.

CORNETT, M. M., MCNUTT, J. J., STRAHAN, P. E., and TEHRANIAN, H. "Liquidity risk management and credit supply in the financial crisis". Journal of Financial Economics, 101, 2011, 297-312.

DEMARZO, P. "The pooling and tranching of securities: A model of informed intermediation" Review of Financial Studies, 18, 2005, 1-35

DEMARZO, P and DUFFIE, D. "A liquidity-based model of security design". Econometrica, 67, 1999, 65-99

DEMYANYK, Y., and VAN HEMERT, O. "Understanding the subprime mortgage crisis". Review of Financial Studies, 24, 2011. 1848-1880.

DIAMOND, D. and RAJAN, R. "Liquidity risk, liquidity creation and financial fragility: A theory of banking". Journal of Political Economy, 91, 2001, 401-419.

DRUCKER, S. and PURI, M. "On Loan Sales, Loan Contracting, and Lending Relationships". Review of Financial Studies, 22, 2009, 2835-2872.

FAMA, E., and FRENCH, K. "Financing Decisions: Who Issues Stock?", Journal of Financial Economics, 76, 2005, 549-82.

FLANNERY, M. "Debt maturity and the deadweight costs of leverage: Optimally financing banking firms", American Economic Review, 84, 1994, 320-331

FRANK, M.Z. and GOYAL, V.K. "Trade-off and pecking order theories of debt" in B.E.Eckbo, Ed., Handbook of Corporate Finance: Empirical Corporate Finance, Vol. 2, Handbook of Finance Series, Amsterdam, Elsevier/North-Holland, 2008.

- "Testing the pecking order theory of capital structure", Journal of Financial Economics, 67, 2003, 217-248.

GATEV, E., SCHUERMANN, T and STRAHAN, P. "Managing Bank Liquidity Risk: How Deposit-Loan Synergies Vary with Market Conditions," Review of Financial Studies, 22, 2009, 995-1020

- GATEV, E. and STRAHAN, P. "Banks' advantage in hedging liquidity risk: theory and evidence from the commercial paper market". Journal of Finance, 61, 2006,867-892
- GORTON, G. B. and PENNACCHI, G. G. "Banks and loan sales. Marketing nonmarketable assets", Journal of Monetary Economics, 35, 1995, 389-411.
- GREENBAUM, S.I. "Bank funding modes: Securitization versus deposits", Journal of Banking and Finance, 11, 1987, 379-401.

JIMÉNEZ, G., MIAN, A.R., PEYDRÓ, and J.L., SAURINA, J. "Local versus aggregate lending channels: The effects of securitization on corporate credit supply in Spain". NBER Working Paper 16595, 2010.

KEYS, B., MUKHERJEE, T., SERU, and A., VIG, V. "Did securitization lead to lax screening? Evidence from subprime loans". Quarterly Journal of Economics. 125, 2010, 14-53.

LOUTSKINA, E. and STRAHAN, P.E. "Securitization and the declining impact of bank finance on loan supply: evidence from mortgage acceptance rates" Journal of Finance, 64, 2009, 861-889.

MIAN, A. and SUFI, A. "The consequences of mortgage credit expansion: evidence from the U.S. mortgage default crisis". Quarterly Journal of Economics, 124, 2009, 1449-1496.

MORGAN, D.P. "Rating banks: Risk and uncertainty in an opaque industry". American Economic Review, 92, 2002, 874-888. MYERS, S. "Determinants of corporate borrowing". Journal of Financial Economics, 5, 1977, 147-175.

- "The capital structure puzzle". Journal of Finance, 39., 1984, 575-592.

MYERS, S. and MAJLUF, N. "Corporate financing and investment decisions when firms have information that investors do not have". Journal of Financial Economics, 13, 1984, 187-221.

PAIS RODRIGUEZ, A. "The role of securitization in the capital structure of banks". Working Paper available at SSRN; http://ssrn.com/abstract=928745. 2005.

- Why do depository institutions use securitization?" Journal of Banking Regulation, 10 (3), 2009, 202-214. PARLOUR, C., and PLANTIN, G. "Loans sales and relation banking", Journal of Finance, 63, 2008, 1291-1314 PENNACCHI, G., Park, K and Han, J. "Corporate taxes and securitization", Journal of Finance, 2014, forthcoming.

PENNACCHI, G. G. "Loan sales and the cost of bank capital", Journal of Finance, Vol. 43, 2, 1988, 375-396.

PURNANANDAM, A. "Originate-to-distribute Model and the Subprime Mortgage Crisis". Review of Financial Studies, 24, 2011, 1881-1915

STRAHAN, P. "Bank structure and lending" in Thakor, A.V and Boot, A., Ed., Handbook of Financial Intermediation and Banking, Handbook of Finance Series, Amsterdam, Elsevier/North-Holland, 2008.

SHYAM-SUNDER, L., and MYERS, R. "Testing static tradeoff against pecking order models of capital structure". Journal of Financial Economics, 51, 1999, 219-244.

LOUTSKINA, E. "The role of securitization in bank liquidity and funding management". Journal of Financial Economics, 100 2011 663-684

	Total Number of Banks		Number of Banks that Securitize at <i>t</i>		Perc. Banks that Securitize at $t(\%)$			Balance of Securitization (mill€)				
	Comm.	Savings	Credit	Comm.	Savings	Credit	Comm.	Savings	Credit	Comm.	Savings	Credit
	Banks	Banks	Coop.	Banks	Banks	Coop.	Banks	Banks	Coop.	Banks	Banks	Coop.
1999	72	48	92	7	17	3	9,7	35,4	3,3	10.434	11.182	190
2000	68	46	90	9	13	3	13,2	28,3	3,3	16.067	11.101	343
2001	66	45	88	10	33	11	15,2	73,3	12,5	17.088	21.378	572
2002	61	45	84	10	30	10	16,4	66,7	11,9	30.632	33.150	1.378
2003	57	45	83	11	39	16	19,3	86,7	19,3	48.550	57.486	3.368
2004	54	45	83	20	41	17	37,0	91,1	20,5	80.763	77.472	6.901
2005	52	45	83	20	44	22	38,5	97,8	26,5	115.345	122.122	11.133
2006	51	45	83	22	43	24	43,1	95,6	28,9	161.526	179.870	16.242

Table 2: Balance sheet of the Spanish Banking system

D'11'		ASS	SETS		LIABILITIES				
Billions €	LOANS	INTERBANK	GOV BONDS	REST	DEPOSITS	DEBT	OWN FUNDS	SEC	
1988	156.63	10.70	50.16	23.99	203.25	12.23	26.00	0.00	
1989	178.77	16.07	59.61	22.87	237.76	11.00	28.55	0.00	
1990	203.07	-3.11	68.15	45.90	268.91	10.98	34.13	0.00	
1991	233.32	2.98	55.91	55.79	293.34	13.36	41.30	0.00	
1992	255.01	-2.12	58.47	64.78	316.42	7.76	46.67	5.30	
1993	267.49	11.01	64.67	74.87	348.43	9.41	52.80	7.40	
1994	282.93	-1.29	85.69	77.31	373.00	11.12	53.19	7.32	
1995	303.58	15.82	97.81	75.53	418.09	11.66	54.88	8.10	
1996	332.33	9.20	104.91	76.42	441.93	15.17	57.62	8.14	
1997	380.19	1.39	94.92	81.93	470.34	19.81	60.06	8.22	
1998	435.30	-31.10	93.81	89.30	494.98	21.56	61.71	9.05	
1999	493.03	-35.33	94.35	111.53	538.68	46.64	65.80	12.46	
2000	605.71	-17.72	91.83	108.86	653.11	29.98	74.68	32.15	
2001	648.01	-4.12	99.53	122.49	707.67	36.62	82.55	39.07	
2002	720.14	-5.68	102.30	119.61	743.40	38.12	89.65	65.20	
2003	824.55	-29.94	111.29	137.30	776.18	60.40	97.16	109.45	
2004	972.32	-12.07	93.28	154.45	812.56	108.23	121.94	165.23	
2005	1226.88	-34.57	92.73	199.43	930.60	167.01	138.11	248.75	
2006	1526.40	17.85	74.10	186.26	1066.66	222.72	157.17	358.08	

A. Volumes (Billions of Euros)

B. Percentages of Total Assets

D C		ASS	SETS			LIABIL	ITIES	
Assets	LOANS	INTERBANK	GOV BONDS	REST	DEPOSITS	DEBT	OWN FUNDS	SEC
1988	64.87	4.43	20.77	9.93	84.17	5.06	10.77	0.00
1989	64.47	5.79	21.49	8.25	85.74	3.97	10.30	0.00
1990	64.67	-0.99	21.70	14.62	85.63	3.50	10.87	0.00
1991	67.05	0.86	16.07	16.03	84.29	3.84	11.87	0.00
1992	67.80	-0.56	15.55	17.22	84.12	2.06	12.41	1.41
1993	63.99	2.63	15.47	17.91	83.35	2.25	12.63	1.77
1994	63.63	-0.29	19.27	17.39	83.89	2.50	11.96	1.65
1995	61.61	3.21	19.85	15.33	84.85	2.37	11.14	1.64
1996	63.56	1.76	20.07	14.62	84.52	2.90	11.02	1.56
1997	68.08	0.25	17.00	14.67	84.23	3.55	10.76	1.47
1998	74.12	-5.30	15.97	15.20	84.28	3.67	10.51	1.54
1999	74.30	-5.32	14.22	16.81	81.18	7.03	9.92	1.88
2000	76.80	-2.25	11.64	13.80	82.68	3.80	9.45	4.07
2001	74.84	-0.48	11.49	14.15	81.73	4.23	9.53	4.51
2002	76.91	-0.61	10.92	12.77	79.39	4.07	9.57	6.96
2003	79.04	-2.87	10.67	13.16	74.40	5.79	9.31	10.49
2004	80.49	-1.00	7.72	12.79	67.27	8.96	10.09	13.68
2005	82.65	-2.33	6.25	13.43	62.69	11.25	9.30	16.76
2006	84.58	0.99	4.11	10.32	59.11	12.34	8.71	19.84

Table 3. Estimation of the relation between credit and deposit growth

	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)
	Total	Banks	Total	Banks	Securitizi	ing Banks	Non-Securitiz	ing Banks
	1988	-1991	2003	8-2006	2003	-2006	2003-20)06
Deposit Growth	0.48 ***	0.35 ***	0.29 ***	0.19 ***	0.30 ***	0.27 ***	0.47 ***	0.23 ***
	(0.08)	(0.04)	(0.08)	(0.04)	(0.07)	(0.04)	(0.10)	(0.08)
GDP Growth	-3.60 ***	-0.73	4.71 ***	4.59 ***	6.21 ***	6.18 ***	6.04 *	6.56 **
	(1.07)	(0.78)	(1.25)	(1.04)	(1.70)	(1.50)	(3.57)	(3.33)
Intercept	0.23 ***	0.16 ***	-0.01	0.00	-0.04	-0.04	-0.12	-0.12
	(0.04)	(0.33)	(0.03)	(0.001)	(0.06)	(0.06)	(0.125)	(0.12)
Fixed Effects	No	Yes	No	Yes	No	Yes	No	Yes
No. of Observ	578	578	743	743	433	433	307	307

Note. Credit Growth is the dependent variable in all the estimations. Symbols: p<0.01 = ***, p<0.05 = **, p<0.1 = *. Standard errors are in parentheses.

Table 4. Probit estimation of	the decision	to	securitize
-------------------------------	--------------	----	------------

	(1)	(2)	(3)	(4)	(5)	(6)
	1(Securitiz	ed 99-06)	1(Securitiz	xed 99-02)	Panel Estimation	
Financial Cost Proxies						
Dep/Loans	-0.226 ** (0.114)	-0.009 (0.053)	-0.247 ** (0.121)	-0.014	-0.059 (0.037)	-0.030
Interbank/ Loans	-0.491	(0.055)	-0.472 (0.429)	(0.055)	-0.358 *** (0.121)	(0.025)
Debt / Loans	0.421 (1.359)		0.391 (1.448)		-0.074 (0.127)	
Equity / Loans	4.265 (2.863)		2.346 (1.862)		-0.448 (0.567)	
Concentration		-0.843 (0.702)		-0.825 (0.607)		-0.221 (0.143)
Liquidity Proxies						
Past Profitability / Loans	-4.770 (9.354)	1.925 (6.418)	-3.903 (9.758)	3.374 (6.488)	-1.635 (1.721)	-2.261 (1.726)
Liquidity / Loans	-1.309 * (0.731)	-1.398 *** (0.477)	-1.347 * (0.744)	-1.532 *** (0.534)	-0.561 *** (0.180)	-0.609 *** (0.153)
Growth Proxies						
Projected Loan Growth	23.485 *** (5.539)	18.393 *** (5.730)	25.961 *** (5.512)	21.356 *** (5.665)	2.967 ** (1.206)	2.748 ** (1.190)
Access to Markets		× ,	× ,	~ /		· · ·
Savings	0.576 ***	0.489 *** (0.120)	0.629 *** (0.131)	0.545 *** (0.126)	0.392 ***	0.449 *** (0.075)
Соор	0.695 ***	0.661 ***	0.738 ***	0.706 ***	0.200 ***	0.299 ***
In Assets	0.288 *** (0.048)	0.201 ***	0.307 ***	0.216 ***	0.122 ***	0.123 *** (0.017)
Bank Control Variables		. ,	. ,	. ,	. ,	
Npl	0.293	0.292	0.355	0.319	9.210 (5.616)	9.649 * (5.778)
RegCap	0.001	-0.141 (0.261)	0.006 (0.282)	-0.124 (0.270)	0.009	(0.017)
Mortg/Loans	-0.572 (0.417)	-0.428 (0.393)	-0.645 (0.439)	-0.478 (0.416)	-0.131 (0.128)	-0.143 (0.137)
No. of Observations	195	195	202	202	1369	1369

Note. (1), (2)= The dependent variable is a dummy that takes the value of 1 if the bank has securitized at least once between 1999 and 2006 and zero otherwise. The explanatory variables refer to the value in 1999. (3), (4)=The dependent variable takes the value of 1 if the bank has securitized at least once between 1999 and 2002 and zero otherwise. The explanatory variables refer to the value in 1999. (5), (6)=The dependent variable takes the value of 1 if the bank has securitized in year *t* and zero otherwise; the estimation includes the time-dummy variables. The explanatory variables refer to the value in *t*-1. The robust standard errors corrected for clustering at the firm and bank level are in parentheses. Symbols: p<0.01 = ***, p<0.05 = **, p<0.1 = *. Standard errors are in parentheses.

(1)	(2)	(3)	(4)	(5)

Table 5. Tobit estimation of the amount securitized

	(1)	(2)	(3)	(4)	(5)	(6)
	1(Securitiz	ed 99-06)	1(Securitiz	ed 99-02)	Panel Estimation	
Financial Cost Proxies						
Dep/Loans	-0.056	-0.014	-0.024	-0.010	-0.016 **	-0.011 **
Interbank/ Loans	(0.069) -0.437 ***	(0.016)	(0.025) -0.166 **	(0.013)	(0.008) -0.052 *	(0.005)
Debt / Loans	(0.142) -0.171 (0.475)		(0.084) -0.062 (0.367)		(0.028) -0.023 (0.036)	
Equity / Loans	-0.844 ** (0.414)		-0.589 * (0.317)		0.078	
Concentration		-0.060 (0.099)		-0.113 (0.077)		-0.009 (0.022)
Liquidity Proxies						
Past Profitability / Loans	0.979 (3.627)	-0.649 (2.496)	1.541 (2.801)	0.097	-0.192 (0.332)	-0.016 (0.365)
Liquidity / Loans	-0.209	-0.323 ** (0.143)	-0.112 (0.149)	-0.164	-0.141 ** (0.055)	-0.138 *** (0.042)
Growth Proxies		()		()	()	
Projected Loan Growth	10.091 ** (4.294)	8.964 ** (4.437)	7.187 ** (3.506)	6.620 * (3.540)	1.015 *** (0.372)	0.978 *** (0.370)
Access to Markets		(,	()		(,	()
Savings	0.062 **	0.110 *** (0.031)	0.057 **	0.066 **	0.038 ***	0.045 ***
Соор	0.032 (0.035)	0.106 ***	0.022 (0.027)	0.057 **	0.012 (0.014)	0.023 (0.014)
In Assets	0.029 ***	0.039 ***	0.025 ***	0.027 ***	0.017 ***	0.016 ***
Bank Control Variables			. ,		. ,	. ,
Npl	0.158	0.164	0.158	0.152	3.176	3.172
RegCap	-0.079	-0.091	-0.020 (0.062)	-0.017	0.006	(1.952) 0.004 (0.008)
Mortg/Loans	0.050 (0.079)	0.065 (0.089)	-0.004 (0.074)	0.001 (0.071)	-0.015 (0.029)	-0.018 (0.031)
No. of Observations	195	195	202	202	1369	1369

Note. (1), (2)= The dependent variable is the ratio of the amount securitized during the period 1999-2006 with respect to assets in 2006 if the bank has securitized and zero otherwise. All of the estimations are robust to heteroskedasticity, and the standard errors are clustered at the bank level. The explanatory variables refer to the value in 1999. (3), (4)= The dependent variable is the ratio of the amount securitized during the period 1999-2002 with respect to assets in 2002 if the bank has securitized and zero otherwise. The explanatory variables refer to the value in 1999. (5), (6)=The dependent variable is the amount securitized in year *t* with respect to assets in *t* and zero otherwise; the estimation includes the time-dummy variables. The explanatory variables refer to the value in *t*-1. The robust standard errors corrected for clustering at the firm and bank level are in parentheses. Symbols: p<0.01 = ***, p<0.05=**, p<0.1 = *. Standard errors are in parentheses.

	(1)	(2)	(3)	(4)	
	Expor	nential	Weibull		
Financial Cost Proxies					
Dep/Loans	0.994 *	0.996	0.994 *	0.996	
Interbank/ Loans	0.983	(-1.40)	0.980	(-1.44)	
Debt / Loans	(-1.41) 1.005 (0.23)		(-1.62) 1.000		
Equity / Loans	(0.25) 0.985 (-0.38)		(-0.01) 0.965 (-0.84)		
Concentration	(-0.58)	0.989	(-0.0+)	0.984 *	
Liquidity Proxies		(-1.17)		(-1.01)	
Past Profitability / Loans	0.995	1.006	0.999	0.985	
Liquidity / Loans	1.009	(0.07)	(-0.01)	(-0.13) 1.009801 (1.02)	
Growth Proxies	(0.90)	(0.75)	(1.10)	(1.03)	
Projected Loan Growth	1.785 ***	1.710 *** (3.84)	1.928 ***	1.833 *** (4.09)	
Access to Markets	(4.07)	(3.04)	(4.55)	(4.07)	
Savings	2.865 *** (2.90)	3.123 *** (3.18)	3.987 *** (3.48)	4.635 ***	
Coop	3.073 ***	4.011 ***	3.834 ***	5.356 *** (4.75)	
In Assets	(5.67) 1.596 ***	(4.07) 1.574 ***	(5.50) 1.819 ***	1.778 ***	
Bank Control Variables	(5.02)	(3.34)	(0.0))	(0.50)	
Npl	0.998	0.998	0.997	0.997 (-0.78)	
RegCap	(-0.50) 1.643 (1.10)	(-0.50) 1.855 (1.48)	(-0.08) 1.298 (0.55)	(-0.78) 1.787 (1.33)	
Mortg/Loans	1.008 (0.85)	1.009 (0.98)	1.009 (0.94)	1.010 (1.00)	
р			1.903 ***	1.912 ***	
No. of Observations	211	211	211	211	

Table 6. Duration Model: Number of years from 1998 to securitization

Note. (1), (2) = The exponential model, hazard rate is constant over time (3), (4)= Weibull model, hazard rate is monotonic if p#1. Symbols: p<0.01 = ***, p<0.05 = **, p<0.1 = *. The results are presented in the form of exponential coefficients, that is, e^{β} because they can be directly interpreted as the increases in the baseline hazard rate. *t*-ratios in parentheses.

		Dependen	t variable: SEC $_t$	
	(1)	(2)	(3)	(4)
FD	0.560 ***	0.571 **	** 0.559 ***	0.573 ***
	(0.063)	(0.105)	(0.064)	(0.107)
FD ·Savings		-0.025		-0.031
		(0.133)		(0.136)
FD ·Coop		0.216 **	k	0.214 **
		(0.106)		(0.108)
$FD \cdot Small$			0.260 ***	-0.263
			(0.092)	(0.285)
FD ·Savings ·Small				0.528 *
C C				(0.309)
FD ·Coop ·Small				-0.113
-				(0.394)
Savings		124,249 *		162,538 *
-		(65,645)		(90,510)
Coop		34,257		44,864
-		(35,324)		(46,704)
Small			-23,701	44,917
			(27,768)	(46,696)
Savings·Small				-159,200 *
-				(90,750)
<i>Coop</i> ·Small				-45,328
-				(46,714)
Intercept		-34,353	22,613	-44,764
-		(35,311)	(27,740)	(46,687)
R^2	85.04	85.38	85.05	85.48
No. of Observ	813	813	813	813

Note. *SEC* is the volume of securitized assets issued by a bank at *t*, and *FD* is the financial deficit of the bank defined in [1]; *Small* is a dummy variable that identifies banks with assets below the 30th percentile of the asset distribution of banks of the same legal nature; *Savings* and *Coop* are dummy variables that identify the savings banks and credit cooperatives, respectively. **OLS** are estimations with the standard errors clustered at the bank level. Symbols: p<0.01 = ***, p<0.05 = ***, p<0.1 = *. Standard errors in parentheses.

Figure 1. Evolution of Banks' Capital Structure



Note. The plot ASSETS displays the relative weights of *LOANS*, *GOV BONDS*, *INTERBANK* and *REST* with respect to total bank assets. The plot LIABILITIES displays the relative weights of *SEC*, *OWN FUNDS*, *DEBT* and *DEPOSITS* with respect to total bank liabilities.

Total Banks

Figure 2. Evolution of Banks' Capital Structure.

2A. Banks that do securitize











LIABILITIES



Note. The plot ASSETS displays the relative weights of *LOANS*, *GOV BONDS*, *INTERBANK* and *REST* with respect to total bank assets. The plot LIABILITIES displays the relative weights of *SEC*, *OWN FUNDS*, *DEBT* and *DEPOSITS* with respect to total bank liabilities.

Figure 3. Evolution of growth rates of loans and deposits.



Total Spanish Banks

Figure 4. Density of the prediction of the number of years until securitization



Total Spanish Banks

APPENDIX. Homogenization of the concepts in the variables LOANS and DEPOSITS

This paper gathers different items of the asset and liability sides of the balance sheet in the following aggregate concepts:

Assets = Loans + Net Interbank + Government Debt + Others (net)

Liabilities = Own Funds + Securitization + Deposits + Debt

Most of the items (net interbank, government debt, own funds, debt, securitization) are obtained directly from the information in the balance sheet. However, loans and deposits require certain adjustments.

For loans, adjustments are made to address a regulatory change introduced at the end of 2004 (CBE 4/2004). Prior to 2004, CBE 4/1991 established that securitized loans were to be written off the balance sheets of banks. With the new regulation in CBE 4/2004, banks could only write off a securitized loan if securitization implied an effective transfer of the risk of that loan. Additionally, the new regulation introduced in 2004 had a retroactive effect, and it obliged banks to include again in their balance sheets those loans securitized in the past that did not comply with the new criteria of risk transfer. As a consequence, there was a break in the series of outstanding loans in year 2005, with approximately 90% of the securitized loans made in the past returning to banks' balance sheets. In order to homogenize these series, we adopt the criterion of the CBE 4/2004. We compute the percentage of off-balance sheet loans that returns to balance sheets in 2005 and extend this percentage to the previous years of the sample (1999-2004). Therefore, the variable *LOAN* will be equal to the accounting item for total loans from 2005 onwards. *LOAN* will include both total loans in the balance sheet and the computed proportion of securitized assets before that year.

For deposits, Spanish accounting regulation establishes that the liability counterparty of an operation of securitization is accounted for in the deposit item of the balance sheet. However, we consider a more refined definition of deposits (i.e., funds collected from consumers in the retail business of banks) and separate them from securitization. To accomplish this, we define *DEPOSITS* as the difference between the deposit item of the balance sheet and the outstanding amount of securitized assets *SEC*.

BANCO DE ESPAÑA PUBLICATIONS

WORKING PAPERS

- 1301 JAMES COSTAIN and ANTON NAKOV: Logit price dynamics.
- 1302 MIGUEL GARCÍA-POSADA: Insolvency institutions and efficiency: the Spanish case.
- 1303 MIGUEL GARCÍA-POSADA and JUAN S. MORA-SANGUINETTI: Firm size and judicial efficacy: evidence for the new civil procedures in Spain.
- 1304 MAXIMO CAMACHO and GABRIEL PEREZ-QUIROS: Commodity prices and the business cycle in Latin America: living and dying by commodities?
- 1305 CARLOS PÉREZ MONTES: Estimation of regulatory credit risk models.
- 1306 FERNANDO LÓPEZ VICENTE: The effect of foreclosure regulation: evidence for the US mortgage market at state level.
- 1307 ENRIQUE MORAL-BENITO and LUIS SERVEN: Testing weak exogeneity in cointegrated panels.
- 1308 EMMA BERENGUER, RICARDO GIMENO and JUAN M. NAVE: Term structure estimation, liquidity-induced heteroskedasticity and the price of liquidity risk.
- 1309 PABLO HERNÁNDEZ DE COS and ENRIQUE MORAL-BENITO: Fiscal multipliers in turbulent times: the case of Spain.
- 1310 SAMUEL HURTADO: DSGE models and the Lucas critique.
- 1311 HENRIQUE S. BASSO and JAMES COSTAIN: Fiscal delegation in a monetary union with decentralized public spending.
- 1312 MAITE BLÁZQUEZ CUESTA and SANTIAGO BUDRÍA: Does income deprivation affect people's mental well-being?
- 1313 ENRIQUE ALBEROLA, ÁNGEL ESTRADA and DANIEL SANTABÁRBARA: Growth beyond imbalances. Sustainable growth rates and output gap reassessment.
- 1314 CARMEN BROTO and GABRIEL PEREZ-QUIROS: Disentangling contagion among sovereign CDS spreads during the European debt crisis.
- 1315 MIGUEL GARCÍA-POSADA and JUAN S. MORA-SANGUINETTI: Are there alternatives to bankruptcy? A study of small business distress in Spain.
- 1316 ROBERTO RAMOS and ENRIQUE MORAL-BENITO: Agglomeration matters for trade.
- 1317 LAURA HOSPIDO and GEMA ZAMARRO: Retirement patterns of couples in Europe.
- 1318 MAXIMO CAMACHO, GABRIEL PEREZ-QUIROS and PILAR PONCELA: Short-term forecasting for empirical economists. A survey of the recently proposed algorithms.
- 1319 CARLOS PÉREZ MONTES: The impact of interbank and public debt markets on the competition for bank deposits.
- 1320 OLYMPIA BOVER, JOSE MARIA CASADO, SONIA COSTA, PHILIP DU CAJU, YVONNE MCCARTHY, EVA SIERMINSKA, PANAGIOTA TZAMOURANI, ERNESTO VILLANUEVA and TIBOR ZAVADIL: The distribution of debt across euro area countries: the role of Individual characteristics, institutions and credit conditions.
- 1321 BRINDUSA ANGHEL, SARA DE LA RICA and AITOR LACUESTA: Employment polarisation in Spain over the course of the 1997-2012 cycle.
- 1322 RODOLFO G. CAMPOS and ILIANA REGGIO: Measurement error in imputation procedures.
- 1323 PABLO BURRIEL and MARÍA ISABEL GARCÍA-BELMONTE: Meeting our D€STINY. A Disaggregated €uro area Short Term Indicator model to forecast GDP (Y) growth.
- 1401 TERESA SASTRE and FRANCESCA VIANI: Countries' safety and competitiveness, and the estimation of current account misalignments.
- 1402 FERNANDO BRONER, ALBERTO MARTIN, AITOR ERCE and JAUME VENTURA: Sovereign debt markets in turbulent times: creditor discrimination and crowding-out effects.
- 1403 JAVIER J. PÉREZ and ROCÍO PRIETO: The structure of sub-national public debt: liquidity vs credit risks.
- 1404 BING XU, ADRIAN VAN RIXTEL and MICHIEL VAN LEUVENSTEIJN: Measuring bank competition in China: a comparison of new versus conventional approaches applied to loan markets.
- 1405 MIGUEL GARCÍA-POSADA and JUAN S. MORA-SANGUINETTI: Entrepreneurship and enforcement institutions: disaggregated evidence for Spain.
- 1406 MARIYA HAKE, FERNANDO LÓPEZ-VICENTE and LUIS MOLINA: Do the drivers of loan dollarisation differ between CESEE and Latin America? A meta-analysis.
- 1407 JOSÉ MANUEL MONTERO and ALBERTO URTASUN: Price-cost mark-ups in the Spanish economy: a microeconomic perspective.
- 1408 FRANCISCO DE CASTRO, FRANCISCO MARTÍ, ANTONIO MONTESINOS, JAVIER J. PÉREZ and A. JESÚS SÁNCHEZ-FUENTES: Fiscal policies in Spain: main stylised facts revisited.

- 1409 MARÍA J. NIETO: Third-country relations in the Directive establishing a framework for the recovery and resolution of credit institutions.
- 1410 ÓSCAR ARCE and SERGIO MAYORDOMO: Short-sale constraints and financial stability: evidence from the Spanish market.
- 1411 RODOLFO G. CAMPOS and ILIANA REGGIO: Consumption in the shadow of unemployment.
- 1412 PAUL EHLING and DAVID HAUSHALTER: When does cash matter? Evidence for private firms.
- 1413 PAUL EHLING and CHRISTIAN HEYERDAHL-LARSEN: Correlations.
- 1414 IRINA BALTEANU and AITOR ERCE: Banking crises and sovereign defaults in emerging markets: exploring the links.
- 1415 ÁNGEL ESTRADA, DANIEL GARROTE, EVA VALDEOLIVAS and JAVIER VALLÉS: Household debt and uncertainty: private consumption after the Great Recession.
- 1416 DIEGO J. PEDREGAL, JAVIER J. PÉREZ and A. JESÚS SÁNCHEZ-FUENTES: A toolkit to strengthen government budget surveillance.
- 1417 J. IGNACIO CONDE-RUIZ, and CLARA I. GONZÁLEZ: From Bismarck to Beveridge: the other pension reform in Spain.
- 1418 PABLO HERNÁNDEZ DE COS, GERRIT B. KOESTER, ENRIQUE MORAL-BENITO and CHRISTIANE NICKEL: Signalling fiscal stress in the euro area: a country-specific early warning system.
- 1419 MIGUEL ALMUNIA and DAVID LÓPEZ-RODRÍGUEZ: Heterogeneous responses to effective tax enforcement: evidence from Spanish firms.
- 1420 ALFONSO R. SÁNCHEZ: The automatic adjustment of pension expenditures in Spain: an evaluation of the 2013 pension reform.
- 1421 JAVIER ANDRÉS, ÓSCAR ARCE and CARLOS THOMAS: Structural reforms in a debt overhang.
- 1422 LAURA HOSPIDO and ENRIQUE MORAL-BENITO: The public sector wage premium in Spain: evidence from longitudinal administrative data.
- 1423 MARÍA DOLORES GADEA-RIVAS, ANA GÓMEZ-LOSCOS and GABRIEL PÉREZ-QUIRÓS: The Two Greatest. Great Recession vs. Great Moderation.
- 1424 ENRIQUE MORAL-BENITO and OLIVER ROEHN: The impact of financial (de)regulation on current account balances.
- 1425 MÁXIMO CAMACHO and JAIME MARTÍNEZ-MARTÍN: Real-time forecasting US GDP from small-scale factor models.
- 1426 ALFREDO MARTÍN OLIVER, SONIA RUANO PARDO and VICENTE SALAS FUMÁS: Productivity and welfare: an application to the Spanish banking industry.
- 1427 JAVIER ANDRÉS and PABLO BURRIEL: Inflation dynamics in a model with firm entry and (some) heterogeneity.
- 1428 CARMEN BROTO and LUIS MOLINA: Sovereign ratings and their asymmetric response to fundamentals.
- 1429 JUAN ÁNGEL GARCÍA and RICARDO GIMENO: Flight-to-liquidity flows in the euro area sovereign debt crisis.
- 1430 ANDRÈ LEMELIN, FERNANDO RUBIERA-MOROLLÓN and ANA GÓMEZ-LOSCOS: Measuring urban agglomeration. A refoundation of the mean city-population size index.
- 1431 LUIS DÍEZ-CATALÁN and ERNESTO VILLANUEVA: Contract staggering and unemployment during the Great Recession: evidence from Spain.
- 1501 LAURA HOSPIDO and EVA MORENO-GALBIS: The Spanish productivity puzzle in the Great Recession.
- 1502 LAURA HOSPIDO, ERNESTO VILLANUEVA and GEMA ZAMARRO: *Finance for all:* the impact of financial literacy training in compulsory secondary education in Spain.
- 1503 MARIO IZQUIERDO, JUAN F. JIMENO and AITOR LACUESTA: Spain: from immigration to emigration?
- 1504 PAULINO FONT, MARIO IZQUIERDO and SERGIO PUENTE: Real wage responsiveness to unemployment in Spain: asymmetries along the business cycle.
- 1505 JUAN S. MORA-SANGUINETTI and NUNO GAROUPA: Litigation in Spain 2001-2010: Exploring the market for legal services.
- 1506 ANDRES ALMAZAN, ALFREDO MARTÍN-OLIVER and JESÚS SAURINA: Securitization and banks' capital structure.

BANCODE ESPAÑA

Eurosistema